Author’s response to reviews

Title: Attentional bias modification in reducing test anxiety vulnerability: A randomized controlled trial

Authors:

Wenpeng Cai (wpcai@smmu.edu.cn)
Yu Pan (531828485@qq.com)
Huangyangzi Chai (yokochyz@163.com)
Yi Cui (cuiyi210@163.com)
Jin Yan (yanjingk@qq.com)
Wei Dong (sophiedongwei@163.com)
Guanghui Deng (bfbedu@126.com)

Version: 5 Date: 26 Oct 2017

Author’s response to reviews:

Dear Editor,

Thank you very much for your careful work. We have studied comments carefully and tried our best to make corrections which we hope meet with approval. The main corrections in the paper and responses to Editor are as following:

1. Please rename the Introduction heading in the main text of the manuscript so that it appears as Background.

Response: Background was used to replace Introduction in this manuscript (Page 3).

2. As indicated by our submission guidelines, please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.

Response: Thank you for your suggestion. Conclusions heading was included and placed before the last paragraph of the original Discussion section (Page 16).
3. Thank you for including a Funding heading for the Declarations section. Please also state the roles of the funding bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: The funders had no role in study design, data collection and analysis, or writing the manuscript. This statement was added in the Funding section (Page 17).

4. Please provide a section after References where you provide the figure legends for your figure files. For each figure, the following information should be provided:

- Figure number

- Short title of figure in sentence case (required)

- Detailed legend (optional)

Response: The figure number and short title were provided after References (Page 21).

Thank you very much for your consideration.

Best regards!

Yours sincerely,

Guanghui Deng

E-mail: bfbedu@126.com